Author’s response to reviews

Title: Barriers and Facilitators to Evidence Based Care of Type 2 Diabetes Patients: experiences of General Practitioners Participating to a Quality Improvement Program.

Authors:

Geert Goderis (geert.goderis@skynet.be)
Liesbeth Borgermans (Liesbeth.borgermans@med.kuleuven.be)
Chantal Mathieu (chantal.mathieu@med.kuleuven.be)
Carine Van Den Broeke (Carine.VanDenBroeke@med.kuleuven.be)
Karen Hannes (Karen.hannes@med.kuleuven.be)
Jan Heyrman (Jan.heyrman@med.kuleuven.be)
Richard Grol (R.Grol@kwazo.umcn.nl)

Version: 5 Date: 12 July 2009

Author’s response to reviews:

Dear Pr. Eccles,

I hereby submit the revised version of the manuscript. I made all proposed changes and I accepted all proposed changes, except but one. The last one (final paragraph of the discussion): I changed "In summary" to "Finally". I also answered to the comments and made the necessary changes. One change is a little bit odd (first page, authors description): as asked I wrote down the cities, but since all mentioned institutions contain the city names (Leuven, Nijmegen), cities are mentioned twice... Since it is essentially a matter of editing, I let you decide whether or not this double mention can stay or not.

Many thanks for your consideration,

Sincerely yours

Dr. Goderis